# SJSU SAN JOSÉ STATE UNIVERSITY



# Influence of Level 1 and Level 2 Automated Vehicles on Fatal Crashes and Fatal Crash Occurrence

Hardik Gajera, MTech Srinivas S. Pulugurtha, PhD, PE, FASCE Sonu Mathew, PhD



#### MINETA TRANSPORTATION INSTITUTE

# Mineta Transportation Institute

Founded in 1991, the Mineta Transportation Institute (MTI), an organized research and training unit in partnership with the Lucas College and Graduate School of Business at San José State University (SJSU), increases mobility for all by improving the safety, efficiency, accessibility, and convenience of our nation's transportation system. Through research, education, workforce development, and technology transfer, we help create a connected world. MTI leads the <u>Mineta Consortium for Transportation Mobility</u> (MCTM) funded by the U.S. Department of Transportation and the <u>California State University Transportation Consortium</u> (CSUTC) funded by the State of California through Senate Bill 1. MTI focuses on three primary responsibilities:

#### Research

MTI conducts multi-disciplinary research focused on surface transportation that contributes to effective decision making. Research areas include: active transportation; planning and policy; security and counterterrorism; sustainable transportation and land use; transit and passenger rail; transportation engineering; transportation finance; transportation technology; and workforce and labor. MTI research publications undergo expert peer review to ensure the quality of the research.

#### Education and Workforce

To ensure the efficient movement of people and products, we must prepare a new cohort of transportation professionals who are ready to lead a more diverse, inclusive, and equitable transportation industry. To help achieve this, MTI sponsors a suite of workforce development and education opportunities. The Institute supports educational programs offered by the Lucas Graduate School of Business: a Master of Science in Transportation Management, plus graduate certificates that include High-Speed and Intercity Rail Management and Transportation Security Management. These flexible programs offer live online classes so that working transportation professionals can pursue an advanced degree regardless of their location.

#### Information and Technology Transfer

MTI utilizes a diverse array of dissemination methods and media to ensure research results reach those responsible for managing change. These methods include publication, seminars, workshops, websites, social media, webinars, and other technology transfer mechanisms. Additionally, MTI promotes the availability of completed research to professional organizations and works to integrate the research findings into the graduate education program. MTI's extensive collection of transportation-related publications is integrated into San José State University's world-class Martin Luther King, Jr. Library.

#### Disclaimer

The contents of this report reflect the views of the authors, who are responsible for the facts and accuracy of the information presented herein. This document is disseminated in the interest of information exchange. MTI's research is funded, partially or entirely, by grants from the California Department of Transportation, the California State University Office of the Chancellor, the U.S. Department of Homeland Security, and the U.S. Department of Transportation, who assume no liability for the contents or use thereof. This report does not constitute a standard specification, design standard, or regulation.

Report 22-22

# Influence of Level 1 and Level 2 Automated Vehicles on Fatal Crashes and Fatal Crash Occurrence

Hardik Gajera, M.Tech.

Srinivas S. Pulugurtha, Ph.D., P.E., F.ASCE

Sonu Mathew, Ph.D.

June 2022

A publication of the Mineta Transportation Institute Created by Congress in 1991

College of Business San José State University San José, CA 95192-0219

### TECHNICAL REPORT DOCUMENTATION PAGE

1. Report No. 22-22	2. Government Accession No.	3. Recipient's Catalog N	lo.
<b>4. Title and Subtitle</b> Influence of Level 1 and Level 2 Automated Vehicles on Fatal Crashes and Fatal Crash Occurrence		<b>5. Report Date</b> June 2022	
		6. Performing Organization Code	
7. Authors Mr. Hardik Gajera: 0000-0001-9010-1355 Dr. Srinivas Pulugurtha: 0000-0001-7392-7227 Dr. Sonu Mathew: 0000-0003-2263-2749		8. Performing Organiza	tion Report
9. Performing Organization Name and Address Mineta Transportation Institute College of Business San José State University San José, CA 95192-0219		10. Work Unit No.	
		11. Contract or Grant No.	
12. Sponsoring Agency Name and Address		13. Type of Report and Period Covered	
U.S. Department of Transportation Office of the Assistant Secretary for Research and Technology University Transportation Centers Program 1200 New Jersey Avenue, SE Washington, DC 20590		14. Sponsoring Agency Code	
13. Supplemental Profes			
<ul> <li>16. Abstract         Connected and automated vehicles (CAVs) are expected to improve safety by gradually reducing human decisions while driving. However, there are still questions on their effectiveness as we transition from almost 0% CAVs to 100% CAVs with different levels of vehicle autonomy. This research focuses on synthesizing literature and identifying risk factors influencing fatal crashes involving level 1 and level 2 CAVs in the United States. Fatal crashes involving level 0 vehicles—ones that are not connected and automated—were compared to minimize unobserved heterogeneity and randomness associated with the influencing risk factors. The research team used the fatal crash data for the years 2016 to 2019 for the analysis. A partial proportionality odds model is developed using crash, road, and vehicle characteristics as the independent variables and the fatal crash involving a vehicle with a specific level of automation as the dependent variable. The results of this research indicate that level 1 and level 2 CAVs are less likely to be involved in a fatal crash at four-way intersections, on two-way routes with wide medians, at nightime, and in poor lighting conditions when compared to level 0 vehicles. However, they are more likely than level 0 vehicles to be involved in a fatal crash with pedestrians and bicyclists. Comparative analysis between vehicles with smart features and other vehicles indicated that pedestrian automatic emergency braking (PAEB) and lane-keeping assistance (LKA) improve the safety by reducing possible collision with a pedestrian and roadside departure, respectively. Contrarily, vehicles with other smart features are still highly likely to be involved in fatal crashes. This research adds to the growing body of literature that will identify potential areas for improvement in the safety of vehicular technologies and road geometry. 17. Key Words Connected and automated vehicles, Fatal Crash, Pedestrian automatic emergency braking, Lane-keeping assistance</li></ul>			
<b>19. Security Classif. (of this report)</b> Unclassified	<b>20. Security Classif. (of this page)</b> Unclassified	<b>21. No. of Pages</b> 59	22. Price

Form DOT F 1700.7 (8-72)

Copyright © 2022

#### by Mineta Transportation Institute

All rights reserved.

#### DOI: 10.31979/mti.2022.2034

Mineta Transportation Institute College of Business San José State University San José, CA 95192-0219

Tel: (408) 924-7560 Fax: (408) 924-7565 Email: mineta-institute@sjsu.edu

transweb.sjsu.edu/research/2034

## Acknowledgments

The authors sincerely thank the Fatality Analysis Reporting System (FARS) for help with the data required for this research.

The authors thank Mr. Chaitanya Bhure for his help with coding to process vehicle identification number (VIN) and identify smart features for analysis and modeling.

Contents
----------

Acknowledgmentsvi
List of Figuresix
List of Tablesx
Executive Summary 1
1. Introduction
1.1 Problem Statement
1.2 Research Objectives
1.3 Organization of the Report5
2. Literature Review
2.1 Factors Influencing Crashes
2.2 Levels of Automation
2.3 Operational and Safety Effects of CAVs7
2.4 Penetration of CAVs
2.5 Limitations of Past Research
3. Methodology 10
3.1 Study Area and Data Collection 10
3.2 Data Processing 10
3.3 Synthesizing the Safety Effects of CAVs 11
3.4 Analysis of Safety Provided by Smart Features 11
4. Study Area, Data Collection, and Processing Methods 12
4.1 Study Area and Data Collection

4.2 Data Processing 12
4.3 Descriptive Statistics
5. Synthesizing the Safety Effects of Level 1 and Level 2 CAVs
5.1 Proportional Odds Model22
5.2 Partial Proportional Odds (PPO) Model
5.3 PPO Model Results
6. Analysis of Effect of Smart Features on Fatal Crash Occurrence
7. Summary & Conclusions
7.1 Summary
7.2 Conclusion
7.3 Limitations and Future Scope
Appendix A
Abbreviations and Acronyms
Endnotes
Bibliography 42
About the Authors

### LIST OF FIGURES

Figure 1. Data Processing Framework	13
Figure 2 Fatal Crashes Involving Level 1 and Level 2 CAVs (2016–2019)	
in the United States	14
Figure 3: Yearly Fatal Crashes in United States Involving Level 1 and Level 2 CAVs	16

### LIST OF TABLES

Table 1: Frequency and Distribution of Variables Related to Persons,
and Vehicles Involved in Crashes
Table 2: Frequency and Distribution of Variables Related to Time,
and Location of Crashes18
Table 3: Frequency and Distribution of Variables Related to Safety,
and Crash Related Factors
Table 4: Analysis of Effects
Table 5: Estimates and Odds Ratios for Models by the Level of Automation 25
Table 6: Comparison of Vehicles with Smart Features and All Vehicles
Table A1. Vehicle Make Details of Level 1 and Level 2 CAVs Involved in Fatal Crashes 35

### **Executive Summary**

Connected and automated vehicles (CAVs) have started to penetrate into the existing transportation system in the past few years. They have the potential to reduce human errors, which is the primary cause of crashes, by either aiding drivers through the use of smart features or by fully eliminating the role of humans in driving at higher levels of automation. The research on CAVs and their potential effects on safety and operations has been analyzed by many researchers using simulation techniques. However, identifying the factors that affect crash occurrence would help answer outstanding questions related to the overall safety effectiveness of CAVs.

The studies on fatal crash occurrence in the past have shown that road geometry, traffic control devices, and the speed of vehicle are common factors causing crashes. With the inclusion of CAVs equipped with smart features, the factors affecting crashes may vary. However, the current literature documents little to no research on the factors influencing fatal crashes and fatal crash occurrence considering real-world crash data of level 1 and level 2 CAVs. Further, the efficiency of smart features tested in laboratories or in controlled environments may vary depending on driving conditions. Thus, there is a need to identify the factors influencing fatal crashes involving vehicles with varying levels of automation and the effectiveness of various smart features in improving safety to proactively plan for infrastructure at higher penetrations, improve safety, and reduce the number of fatal crashes. The objectives of this research, therefore, are (1) to collect and comprehensively evaluate data pertaining to the levels of vehicle automation and what each level entails from a safety perspective, considering selected models/makes, their manufacture year, and specifications; (2) to research the trends in the penetration of level 1 and level 2 CAVs; and (3) to model the effect of level 1 and level 2 CAVs on fatal crashes and fatal crash occurrence.

Fatal crash data for 2016–2019 was obtained from the Fatality Analysis Reporting System (FARS) database and vehicle identification numbers (VINs) of all vehicles involved in crashes were extracted. Using the VINs, smart features in each vehicle involved in a crash were retrieved from the National Highway Traffic Safety Administration (NHTSA) database by creating a tool in Python. Crash related datafiles and data of smart features were combined to form a complete dataset for modeling. Using the Society of Automobile Engineers (SAE) International levels of automation, each vehicle was classified into a level of automation based on the smart features in the vehicle.

For the purpose of comparison of crashes involving level 1 and level 2 CAVs with level 0 vehicles, three nearest fatal crashes involving level 0 vehicles were identified and considered as corresponding samples for level 0 crashes in modeling. A proportional odds (PO) test was carried out to identify the slopes of different independent variables and it was identified that some of the independent variables have unequal (varying) slopes. Thus, a partial proportional odds (PPO) model was developed to identify the factors influencing crashes involving level 1 and level 2 CAVs compared to crashes involving level 0 vehicles. To identify the effectiveness of various smart

features, which are designed to enhance safety for corresponding types of crashes, comparative analysis was carried out between vehicles equipped with features and other vehicles involved in fatal crashes.

The results from the PPO model indicate that level 1 and level 2 CAVs are less likely to be involved in crashes at four-way intersections, on two-way routes with medians, at nighttime, and in conditions with poor lighting compared to level 0 vehicles. In contrast, the CAVs have higher odds of being involved in crashes with non-motorists such as pedestrians and bicyclists compared to level 0 vehicles. The CAVs were also found to be more involved in crashes on one lane routes compared to level 0 vehicles. The results from the comparative analysis indicated that adaptive cruise control (ACC) and forward collision warning system (FCWS) are not efficient in improving safety in case of rear-end collisions. However, vehicles with pedestrian automatic emergency braking (PAEB) and lane-keeping assistance (LKA) are efficient in improving safety by reducing collisions with pedestrians and roadside departures, respectively. The findings and results from this research could be used in identifying the factors affecting fatal crashes involving CAVs, and potential areas for improvement in vehicular technologies as well as road geometry.

### 1. Introduction

In the United States, motor vehicle crashes are one of the top ten causes of death,<sup>1</sup> resulting in congestion and increasing safety concerns. In 2019, over 2.35 million people were injured or disabled, and 36,096 people lost their lives solely because of road crashes.<sup>2</sup> The cost of one fatality related to a motor vehicle crash is \$1,704,000, while the cost of evident injury is \$28,500.<sup>3</sup> According to the Centers for Disease Control and Prevention (CDC), the cost of medical care and productivity losses due to motor vehicle crashes (injuries and fatality) was reported to be more than \$75 billion in the United States.<sup>4</sup> Drivers are at risk of being involved in a crash, regardless of whether they drive the safest vehicle or on familiar roads in normal conditions.

Several factors related to driver errors, such as improper lane changes, excessive speed, and inattentiveness while driving, cause many crashes. According to Traffic Safety Facts from the National Highway Traffic Safety Administration (NHTSA), approximately 94% of crashes in the United States between 2005 and 2007 were caused due to human errors.<sup>5</sup> Connected and automated vehicles (CAVs) are expected to enhance traffic safety and operation by reducing human involvement in various driving tasks with the help of assistance provided by smart features. In the past few years, a plethora of research work focusing on the effects of CAVs on traffic safety, operation, and human involvement have been published by researchers working in transportation. However, level 1 and level 2 CAVs are already penetrating the market, and their effects on crash occurrence needs to be identified.

The smart features engaged in CAVs reduce the involvement of humans in driving and could potentially eliminate human errors. However, driver reliance on these features may also result in inattentiveness while driving. While sitting idle in the driving seat, drivers may use mobile phones or perform other secondary tasks, which could result in cognitive distractions, affecting attention and judgment.<sup>6</sup> Thus, it is also important to examine changes in driver's behavior while driving vehicles equipped with varying levels of automation to determine the potential safety effects of the smart features.<sup>7</sup> Further, the attentiveness of drivers has also been found to vary based on individuals age and gender, as well as road environment.<sup>8</sup>

Investigating the factors affecting crashes involving level 1 and level 2 CAVs provide insights on the involvement of CAVs equipped with smart features designed to enhance safety in certain types of crashes. Furthermore, the findings of this study also help to identify the involvement of CAVs in crashes, which would help manufacturers and practitioners modify existing CAVs and design new policies.

#### 1.1 Problem Statement

Motor vehicles and driver interactions are likely to change significantly in the next few decades, perhaps more than they have in the past. Recent and ongoing advances in vehicle automation

technology have created very high expectations regarding highway performance, safety improvements, and environmental benefits. As human error is the leading cause of road crashes in the United States, CAVs are expected to reduce the number of crashes caused by drivers through the gradual removal of the role of human decisions in driving.

Although the deployment of driving assistance and autopilot features has increased over recent years, fully automated CAVs are not yet a reality, apart from a few test vehicles. Without a high penetration rate, the safety benefits of CAVs may not be maximized. Furthermore, the safety benefits also depend on how heavily the driver relies on driver assistance, autopilot, and other smart features. Additionally, some recent crashes involving vehicles with collision avoidance and autopilot systems, resulting in deaths, indicate that CAVs may not yet be effective all of the time. Potential reasons are related to disengagement and smart features abilities to sense and control the CAV irrespective of the geographic location, geometric configurations, environmental and traffic conditions, and time of day.

Understanding the effect of the transition from no CAVs to level 1 and level 2 CAVs, and ultimately all CAVs, on the overall safety of the transportation system could be more challenging than expected. A comprehensive safety analysis to examine the trends in crashes over time in conjunction with the advancements in vehicle technology is the first step. Such analysis should be complemented with modeling to identify factors associated with level 1 and level 2 CAV crash involvement when compared to non-automated (level 0) vehicles. Therefore, there is a need to analyze crash data and identify factors that play a role in crashes involving level 1 and level 2 CAVs. Due to the potential reduction in driving efforts and the transition to CAVs, the conventional perception of ownership of vehicles can also be influenced over time. Therefore, there is a need to examine the market trends, probable shift to CAVs (mode shift), and projected effect of vehicles with different levels of automation on crashes. This will help in developing a readiness plan to proactively address anticipated safety challenges in future years.

#### 1.2 Research Objectives

The objectives of the proposed research are as follows:

- To collect and comprehensively evaluate data pertaining to the levels of vehicle automation and what each level entails from a safety perspective, considering selected models/makes, their manufacture year, and specifications.
- To research the trends in the penetration of level 1 and level 2 automated vehicles (AVs).
- To model the effect of level 1 and level 2 CAVs on fatal crashes and fatal crash occurrence.

The descriptive statistics from crash data provide an overview of the number of crashes involving level 1 and level 2 CAVs. The PPO model results compare the CAVs with level 0 vehicles and

identify the factors affecting occurrence of fatal crashes involving level 1 and level 2 CAVs. The findings also help in identifying the factors related to vehicle, road geometry, and crash, for implementing policies to improve the safety of the existing transportation system and serve as an overview of current involvement of CAVs in fatal crashes.

#### 1.3 Organization of the Report

The remainder of the report comprises six chapters. Chapter 2 summarizes past literature related to the effects of CAVs on safety, factors affecting fatal crashes, occurrence of fatal crashes, and market penetration trends of CAVs. Further, the effects of CAVs on the occurrence of fatal crashes is summarized along with the limitations of past research. Chapter 3 presents the methodological framework adopted for this research. Chapter 4 describes the study area, data collection, and processing methods along with the descriptive statistics of the data used in this research. Chapter 5 discusses the modeling technique used and the results from the partial proportional odds (PPO) model of level 1 and level 2 CAVs. The summary of vehicles equipped with various smart features and their efficiency in reducing fatal crashes is presented in Chapter 6. Chapter 7 presents the summary of this research, along with conclusions and the scope of future research.

### 2. Literature Review

This chapter presents an overview of past research associated with factors affecting fatal crashes, levels of automation, and the effects of CAVs on traffic safety and operations. Further, additional discussions related to studies on the effect of CAVs on mode choice, trip generation, and penetration of CAVs into the automobile market are presented.

#### 2.1 Factors Influencing Crashes

In the past, many researchers focused on identifying factors influencing crashes, such as the dimension of medians,<sup>9</sup> side traffic barriers,<sup>10</sup> speed limits,<sup>11</sup> road infrastructure,<sup>12</sup> highway class, demographic characteristics<sup>13</sup> and adverse weather conditions.<sup>14,15</sup> Some researchers also evaluated the effects of red-light cameras,<sup>16,17</sup> road surface,<sup>18</sup> intersection type,<sup>19</sup> and annual average daily traffic (AADT) on crashes at intersections.<sup>19,20</sup> These studies are a few examples efforts on identifying the factors related to crashes or using before and after analysis to determine improvements in traffic safety.

In addition to the road and geometry-related factors, vehicle characteristics (smart features, safety standards, size, and type of vehicle) also influence crash injury severity. However, most vehicle safety devices are considered as secondary measures as their existence cannot prevent a vehicle from getting involved in a crash. Their sole purpose is to reduce the effect of a crash on the drivers and passengers. For example, vehicle safety devices such as seatbelts,<sup>21-23</sup> airbags,<sup>22</sup> and antilock braking systems<sup>21</sup> reduce injury severity in a crash.

A report published by the NHTSA shows that more than 94% of all motor vehicle crashes are caused by human error.<sup>2</sup> The causes of human error generally vary for different age groups. Teen drivers (below 20 years of age) generally get involved in crashes because of their immaturity, lack of skills, inexperience, and aggressive driving nature. Thus, the crash rate per miles driven and crash rate per number of license holders are higher for teens than for adults.<sup>24</sup> In contrast, elderly drivers (above 65 years) generally suffer from age or health-related problems which affect their reaction time, ability to divide attention between multiple tasks, and vision,<sup>25</sup> due to which their chances of getting involved in a crash are higher compared to adult drivers.<sup>26</sup> In addition to age, several other factors such as gender, distracted driving, and driving under the influence of alcohol or drugs also influence the likelihood of getting involved in crashes.

#### 2.2 Levels of Automation

CAVs are expected to reduce the number of crashes caused by drivers through the gradual removal of the role of human decisions in driving. CAVs are characterized as smart vehicles that can interact with other vehicles and infrastructure to avoid any possible crashes resulting from human errors like inattentiveness, distracted driving, or aggressive driving. They are driving the existing market because of increasing emphasis towards safety, advancement of the vehicle to vehicle (V2V)

and vehicle to infrastructure (V2I) connectivity, and introduction of the internet of things (IoT) in the automobile industry.

According to the Society of Automotive Engineers (SAE) International, CAVs will integrate on roads using six different automation levels.<sup>27</sup> Vehicles without any smart features are categorized as level 0 (or non-CAVs). Vehicles that the driver could control but with features to assist the driver in handling lateral movement by ensuring that the vehicle stays in the lane or linear movement by controlling the acceleration and braking function qualifies as level 1. Partially automated (level 2) vehicles have automated functions that can control both acceleration and steering, but, driver has to steer, accelerate or break when needed to maintain safety.<sup>27-29</sup>

Level 3 vehicles should perform all driving tasks under limited circumstances such as driving on freeways or straight routes. Simultaneously, the human driver has to control the vehicle at any time, especially when there are multiple lane markings (mostly at intersections). Level 4 and level 5 vehicles are not yet available in automobile markets for vehicle owners. Still, several automobile manufacturers are working on developing such vehicles that have the capabilities of self-driving and require limited to no human efforts in performing driving tasks.<sup>27</sup>

### 2.3 Operational and Safety Effects of CAVs

Several studies have evaluated the effect of CAVs on road operational performance by considering various measures such as the average speed,<sup>30</sup> travel time and travel time reliability,<sup>31-33</sup> number of stops,<sup>34</sup> and delay at nodes or intersections in a particular network.<sup>34</sup> A few researchers have evaluated the safety effects of CAVs using surrogate safety assessment model (SSAM).<sup>35,36</sup> Some researchers identified the effects of smart features of CAVs on safety. Examples include the effects of automated braking system (ABS),<sup>37</sup> lane-keeping assistance (LKA),<sup>38</sup> adaptive cruise control (ACC),<sup>37,39</sup> forward collision warning system (FCWS),<sup>38</sup> and crash avoidance technology<sup>37</sup> on safety. To analyze the effects of vehicles on safety, researchers have developed different parametric and non-parametric models such as negative binomial model,<sup>40</sup> spatial autoregressive model,<sup>40</sup> modified negative binomial regression,<sup>41</sup> multivariate adaptive regression,<sup>40</sup> bootstrap-based binary logistic regression,<sup>42</sup> random parameter models,<sup>43</sup> and intelligent driver models.<sup>44</sup>

Based on the detailed review of previous research studies, microscopic simulation is the most common method adopted to evaluate the safety effects of CAVs.<sup>45</sup> However, accurately calibrated models and appropriate surrogate safety measures are required to improve the degree of reliability of the simulation results.<sup>45</sup> This is not feasible for this study due to the insignificant number of CAVs in use in the transportation system at this time. Although level 1 and level 2 CAVs are available in recent years, only a few studies have used real-world data to identify CAVs safety effectiveness.

There are limited studies comparing the safety effects of CAVs with level 0 vehicles. Researchers have conducted a comparative analysis of the driving potential of human drivers and CAVs using

real world data of crash rate per million vehicle miles travelled.<sup>46</sup> There are potential barriers that CAVs must overcome to eliminate human interaction while performing driving tasks in a realworld scenario. CAVs are considered to be safe and efficient, but are still involved in crashes. The potential reasons they are mostly involved in crashes include disengagement of automated features, false detection of objects, and perception discrepancies.<sup>47,48</sup> Additionally, some recent crashes involving vehicles with collision avoidance and autopilot systems, resulting in deaths, indicate that CAVs may not be yet effective all of the time.

#### 2.4 Penetration of CAVs

The benefits of CAV technology largely depend on a higher market penetration rate. The CAVs are expected to enhance traffic safety and operational performance of a transportation system through reduced reaction times, shorter gaps between vehicles (platooning), and efficient route choices. Market penetration is the percent of CAVs in the total fleet mix. Over the years, market penetration may vary from level 1 (partial automation) to level 5 (full automation). A few studies aimed to predict market penetration over time by analyzing technology trends and travel preferences.<sup>49,50</sup> Lavasani et al. (2016)<sup>49</sup> developed a market penetration model based on the adoption patterns of other technologies such as smartphones and the internet. Assuming CAVs are available in 2025, the study results indicate 7% CAV penetration in 2035 and 75% CAV penetration in 2060 in the United States.<sup>49</sup>

Similarly, Chen et al. (2016) also found that it would take between 12 and 25 years for 75% penetration of CAVs.<sup>50</sup> Kröger et al. (2019) used a vehicle technology diffusion model and presented existing trends and extreme adoption scenarios for the year 2035 in United States and Germany.<sup>51</sup> The penetration of CAVs based on the existing trend scenario was projected to be 10% and 8% for United States and Germany, respectively. However, considering the extreme adoption scenario, the CAVs penetration rate is expected to be higher (38%) in Germany compared to the United States (29%) due to a higher share of luxury cars and quicker fleet turnover. Bansal and Kockelman (2017) surveyed 2,167 Americans using a stated preference survey about CAV technologies.<sup>52</sup> The results suggest that privately-owned CAVs would have 24.8% penetration by 2045, assuming an annual reduction of 5% in the price of a CAV. However, the share jumps to 87.2% if prices decline by 10% every year.

Nieuwenhuijsen et al. (2018), in their study on penetration of CAVs, generated three possible scenarios by varying parameters such as consumer's attitude towards CAVs, economic growth, technological developments, and policies related to CAVs.<sup>53</sup> In the base scenario, varying policies related to CAVs and car-sharing success rates were tested. It was estimated that in 2025, the market share of level 1 and level 2 CAVs would be 21% and 51% respectively. In the other two scenarios, positive attitudes towards CAVs, strong economic growth, and higher technological enhancements were considered, and it was projected that in 2025, market penetration of level 1 and level 2 CAVs would be 8% and 24% under conservative, and 3% and 10% under the progressive scenario.<sup>53</sup>

#### 2.5 Limitations of Past Research

Understanding the effect of the transitions from level 0 to level 1 and some level 2 CAVs to all CAVs on the overall safety of the transportation system could be more challenging than expected. A comprehensive safety analysis to examine the trends in crashes in conjunction with advancements in vehicle technology is the first step. It should be complemented with synthesizing and identifying risk factors associated with fatal crashes involving level 1 and level 2 CAVs when compared to level 0 vehicles. Therefore, this research focuses on bridging this research gap by synthesizing and identifying risk factors influencing fatal crashes involving level 1 and level 2 CAVs when compared to level 0 vehicles. Further, identification of makes of all level 1 and level 2 CAVs involved in fatal crashes would also provide information to the practitioners as well as industrial experts for future vehicular and policy modifications.

While the penetration of CAVs into the market is expected to increase over time, they currently account for hardly 1% of the vehicles using the transportation system. A quick comparison of vehicles involved in fatal crashes indicate that level 1 and level 2 CAVs are involved in ~1.8% of fatal crashes from 2016 to 2019 in the United States. Considering all level 0 vehicles involved in fatal crashes for comparative analysis could skew the research findings. The influence of risk factors on crash involvement could be controlled by using crashes involving level 0 vehicles within the vicinity of level 1 and level 2 CAVs. Such a nearest neighbor-based study design will help to identify and better understand the role of risk factors influencing fatal crashes involving level 1 and level 2 CAVs compared to level 0 vehicles. It will also help minimize the effect of unobserved heterogeneity and randomness associated with the influencing risk factors.

### 3. Methodology

This chapter presents the methodology to identify and compare the factors influencing fatal crashes involving level 1 and level 2 CAVs with level 0 vehicles.

#### 3.1 Study Area and Data Collection

The study area for the research was selected to maximize the number of level 1 and level 2 crashes to assess the effects of spatially varying geometric characteristics. To examine the safety effects of level 1 and level 2 CAVs on the United States' transportation system, samples from all states were considered for analysis in the present research. For identifying the sale and penetration of level 1 and level 2 CAVs, various statistical datasets and the current literature on market penetration was reviewed.

To identify the effect of various levels of automation on fatal crashes, fatal crash data of United States was collected from the Fatality Analysis and Reporting System (FARS) database. Fatal crash data from 2016 to 2019 was considered for the purpose of modeling. Further, to identify the smart features in vehicles involved in fatal crashes, the vehicle identification number (VIN) of all the vehicles was used and data pertaining to smart features was retrieved from the NHTSA database.

#### 3.2 Data Processing

This step involved processing the raw data obtained from FARS and NHTSA databases. The raw FARS data was divided into separate files related to crash, vehicle, pedestrian, and other characteristics. Initially, the data obtained in separate files from 2016 to 2019 was linked using the case ID and year, which were the common fields in all files. Further, the data from each year was combined to form a common dataset. Some of the samples had missing values or not-reported data, and were subsequently removed.

The VIN of all vehicles in the combined dataset was extracted as a separate file and data related to all vehicles based on VIN was obtained using a Python script. A loop was created to identify information of all the vehicles in a single trial. The loop looks up each VIN in the input list, connects with the VIN dataset of NHTSA, and returns the information of all smart features in the form of a list. Finally, the retrieved data including the VINs and information about smart features engaged in vehicle was joined with the FARS dataset.

Level 1 and level 2 vehicles made up only ~1.8 % of the obtained dataset. Further, various regions were identified to have no crashes involving level 1 or level 2 vehicles. Thus, for a more balanced comparison, and to minimize the spatial variance in the geometric characteristics, for every crash involving a level 1 or level 2 vehicle, three nearest neighbors (level 0 crashes) were identified. Samples involving one or more CAVs with different levels of automations were also identified and considered as separate data points in the analysis. Before modeling, the samples with unknown or

unidentified values were eliminated. Chapter 4 provides a detailed overview of the study area, data collection, and processing.

### 3.3 Synthesizing the Safety Effects of CAVs

The factors affecting level 1 and level 2 CAVs may vary due to vehicular characteristics. Thus, both levels of automation were considered as separate categories. A variable indicating the highest level of automation amongst all vehicles involved in a crash was considered as the dependent variable. The partial proportional odds (PPO) model is popular amongst various logistic regression models because it provides flexibility by considering varying slopes of the independent variables. As the level of automation is considered as the dependent variable, the PPO model provides flexibility to model the factors which may not get affected due to level 2 automation compared to level 1 automation with equal slope along with unequal slopes for other variables. Thus, a PPO model was developed using level of automation as a dependent variable and factors related to road geometry, crash and vehicular characteristics, as well as pedestrians and bicyclists involved in crash as independent variables.

#### 3.4 Analysis of Safety Provided by Smart Features

The projected safety benefits of CAVs due to the presence of smart features also needs to be evaluated to identify the pros and cons of the CAVs. To identify the effect of smart features on the occurrence of fatal crashes, vehicles equipped with various smart features were identified using the smart features data obtained from the NHTSA database. Likewise, the purpose for which smart features were designed and the corresponding crash types they could mitigate were also identified. The involvement of vehicles with and without smart features was summarized in Chapter 6 to assess the overall improvement in safety due to presence of smart features.

# 4. Study Area, Data Collection, and Processing Methods

This chapter provides an overview of the study area, data, and data processing. Descriptive statistics are also presented in this chapter.

#### 4.1 Study Area and Data Collection

The data pertaining to fatal crashes in the United States from 2016 to 2019 was obtained from the FARS database. The FARS database contains information related to fatal crashes including information on the vehicles involved in the crashes, pedestrian involvement in the crashes, and other factors. The geometric condition of the road on which each crash occurred, the weather conditions, and time of day are also included in the database. Further, the information related to smart features in vehicles involved in each crash was obtained separately from the NHTSA database using VIN.

#### 4.2 Data Processing

The data related to all the parameters was obtained from the FARS and NHTSA databases. The data processing used for this research is summarized as shown in Figure 1.

Figure 1. Data Processing Framework



The crash database consisted of several files with different crash parameters related to vehicles, pedestrians, drivers, and visibility while driving. All datasets share a common case number, which was used to compile the full dataset used in this research. A Python script was developed to generate a loop including all the VINs, which directly calls the NHTSA database and returns all the information related to smart features such as ACC, LKA (sometimes referred to as lane centering assistance), pedestrian automatic emergency braking (PAEB), and FCWS.

The data obtained from the NHTSA database had information about several features engaged in vehicles. As per the six levels of automation described by SAE International,<sup>27, 28</sup> the vehicles were classified into three different groups: level 0, level 1, and level 2. The vehicles without the LKA or ACC were classified as level 0 vehicles. The vehicles with either LKA or ACC were classified as level 0 vehicles. The vehicles with either LKA or ACC were classified as level 2 CAVs. Information about engagement of features in a crash was not available in the dataset.

The VIN data showed that a limited number of level 1 and level 2 CAVs were involved in fatal crashes before 2016. While not many level 1 and level 2 CAVs were purchased by vehicle owners prior to 2016, the number has increased considerably in recent years. Further, the crash data also

indicated that only 46 out of 52,714 vehicles involved in a crash in 2016 qualified as either level 1 or level 2 automation. Therefore, the crash data of previous years was not considered and only data for the years 2016 to 2019 was used for the analysis. In case of a crash involving both a level 1 and level 2 CAVs, it was considered in both the categories to capture the effect of vehicular characteristics on crash occurrence. Figure 2 shows the location of all crashes involving level 1 and level 2 CAVs in the United States. The total number of fatal crashes involving level 1 or level 2 CAVs was 2,428 (~1.8%) compared to 136,471 (~98.2%) fatal crashes involving level 0 vehicles.

Figure 2. Fatal Crashes Involving Level 1 and Level 2 CAVs (2016-2019) in the United States



The number of crashes involving level 1 and level 2 CAVs increased from 22 and 24, respectively, in 2016 to 537, and 1,062 in 2019. Figure 3 shows the yearly number of fatal crashes involving level 1 and level 2 CAVs. It is noticeable that level 2 crashes are almost double the number of level 1 crashes in 2019 which may be due to the increasing penetration of level 2 CAVs compared to

level 1 CAVs. The involvement of level 1 and level 2 CAVs in fatal crashes also varied based on the make and model of the vehicle. The number of level 1 and level 2 CAVs involved in fatal crashes is summarized in Appendix 1.



Figure 3: Yearly Fatal Crashes in United States Involving Level 1 and Level 2 CAVs

If the sample size is uneven Cover and Hart recommend using nearest neighbors for the analysis.<sup>54</sup> The number of neighbors selected for comparison should be large enough to reduce the chances of biased estimates. Selecting neighbors in proportion to the samples in the reference group (in this research crashes involving level 1 and level 2 CAVs) ensures that points are close enough to provide accurate estimates.<sup>54</sup> This is also driven by past research findings that crashes are spatially correlated.<sup>55,56</sup> Considering fatal crashes involving level 0 vehicles within the vicinity of crashes involving level 1 and level 2 CAVs also minimizes unobserved heterogeneity and randomness associated with influencing risk factors. Therefore, to compare the CAVs with non-CAVs and to identify an appropriate sample size for crashes involving level 0 vehicles corresponding to crashes involving level 1 and level 2 CAVs, the three nearest neighbors from locations of crashes involving level 1 or level 2 CAVs were identified using the 'nearest neighbor' tool in ArcGIS Pro. Further, the samples with null values amongst identified crashes involving level 1 and level 2 CAVs and nearest neighbors involving level 0 vehicles were filtered to remove the samples with not-specific, unknown, or unidentified values as per the FARS database.

#### 4.3 Descriptive Statistics

A descriptive analysis was carried out to compute the frequency distribution amongst the different categories of the independent variables. The frequency and percentage for various categories of variables in the dataset were considered for identifying the suitable technique for analysis. Table 1 shows the descriptive statistics of the crashes involving pedestrian, bicyclists, and vehicles.

Variable	Category	Frequency (%)
	0	10,127 (80.66)
Level of automation	1	863 (6.87)
	2	1,565 (12.47)
	1	3,974 (31.65)
X7 1 · 1 · 1 · 1	2	5,835 (46.48)
Vehicles involved	3	1,728 (13.76)
	>=4	1,018 (8.11)
Dedactrice investored	No	10,841 (86.35)
Pedestrian involved	Yes	1,714 (13.65)
Bicyclist involved	No	12,309 (98.04)
	Yes	246 (1.96)

# Table 1. Frequency and Distribution of Variables Related to Persons,and Vehicles Involved in Crashes

The descriptive statistics show that the majority of crashes involved either 1 or 2 vehicles. However, 8.11% crashes involved more than four vehicles. In addition, pedestrian involvement in fatal crashes is higher compared to the involvement of bicyclists. The proportion of level 1 and level 2 CAVs involved in fatal crashes is 6.87% and 12.47%, respectively.

The variables related to time, location, and functional class of the road on which crashes occurred are shown along with the frequency and percentages in Table 2. They indicate that the majority of the crashes occurred on two lane routes, and at non-junction locations compared to other route types and locations of route. Further, crashes in the urban areas are higher compared to the rural areas. The temporal variation of crashes also indicates that the number of crashes is higher on Friday, and Saturday compared to other days of week. The number of crashes is also higher from 3:00 PM to 9:00 PM compared to other time periods. However, the variation is marginal amongst all the categories of time of day and day of the week due to which relying on descriptive statistics may not provide clear idea about the factors influencing fatal crashes.

The descriptive statistics of factors related to safety, traffic control measures, and other factors is summarized in Table 3. They indicate that normal conditions such as dry road surface, vehicle traveling straight before the crash, and tracking (stable) vehicle before the crash occur with higher frequencies compared to other categories of corresponding variables. The categories of all the variables were observed to identify the ideal conditions as per the past literature and were considered as base categories.

Variable	Category	Frequency (%)
Day of the week	Monday	1,753 (13.96)
	Tuesday	1,540 (12.27)
	Wednesday	1,636 (13.03)
	Thursday	1,778 (14.16)
	Friday	2,022 (16.11)
	Saturday	2,010 (16.01)
	Sunday	1,816 (14.46)
	0:00–2:59 am	1,209 (9.63)
	3:00–5:59 am	1,026 (8.17)
	6:00–8:59 am	1,277 (10.17)
T: C.1 1	9:00–11:59 am	1,333 (10.62)
I ime of the day	12:00–2:59 pm	1,821 (14.5)
	3:00–5:59 pm	2,152 (17.14)
	6:00–8:59 pm	2,011 (16.02)
	9:00–1:59 pm	1,726 (13.75)
A	Rural	4,591 (36.57)
Area type	Urban	7,964 (63.43)
	Interstate	2,121 (16.89)
	Principal arterial—other freeways and expressways	810 (6.45)
	Principal arterial—other	4,400 (35.05)
Functional class	Minor arterial	2,618 (20.85)
	Major collector	1,288 (10.26)
	Minor collector	315 (2.51)
	Local	1,003 (7.99)
	Non-trafficway or driveway access	94 (0.75)
	One lane	172 (1.37)
	Two lanes	6,910 (55.04)
	Three lanes	1,925 (15.33)
Number of lanes	Four lanes	1,705 (13.58)
	Five lanes	1,238 (9.86)
	Six lanes	333 (2.65)
	Seven or more lanes	178 (1.42)
	Acceleration/deceleration lane	16 (0.13)
	Crossover-related	28 (0.22)
Portion of the road on	Driveway access related	512 (4.08)
which crash occurred	Entrance/exit ramp related	236 (1.88)
	Intersection-related	3,837 (30.56)
	Non-junction	7,614 (60.65)

Table 2. Frequency and Distribution of Variables Related to Time and Location of Crashes

Variable	Category	Frequency (%)
	Other location within interchange area	95 (0.76)
	Railway grade crossing	20 (0.16)
	Through road	197 (1.57)
	Three-way intersection	1,263 (10.06)
	Four-way intersection	2,550 (20.31)
Type of intersection	Roundabout, traffic circle, or multiple-way intersection	24 (0.19)
	Not an intersection	8,718 (69.44)
Drassa of seconds as a	No	12,203 (97.2)
Presence of work zone	Yes	352 (2.8)
	One-way trafficway	140 (1.12)
	Two-way, not divided	5,850 (46.59)
	Two-way, not divided with a continuous left-turn lane	851 (6.78)
T	Two-way, divided, positive median barrier	2,492 (19.85)
I rafficway type	Two-way, divided, unprotected (painted > 4 feet) median	469 (3.74)
	Two-way, divided, unprotected median	2,437 (19.41)
	Entrance/exit ramp	222 (1.77)
	Non-trafficway or driveway access	94 (0.75)

Variable	Category	Frequency (%)
<b>X</b> 7.1.1	No	4,801 (38.24)
venicle at fault	Yes	7,754 (61.76)
Hit and run	No	12,355 (98.41)
	Yes	200 (1.59)
D 11	No	10,932 (87.07)
Rollover	Yes	1,623 (12.93)
	Angle	3,336 (26.57)
	Front-to-front	1,918 (15.28)
	Front-to-rear	1,712 (13.64)
	Rear-to-side	39 (0.31)
Manner of collision	Sideswipe—opposite direction	241 (1.92)
	Sideswipe—same direction	378 (3.01)
	The first harmful event was not a collision with a motor vehicle in transport	1,421 (11.32)
	Not a collision with motor vehicle in-transport	3,510 (27.96)
	No damage	254 (2.02)
Damage to the	Minor damage	1,163 (9.26)
vehicle	Functional damage	1,568 (12.49)
	Disabling damage	9,570 (76.22)
	Daylight	6,380 (50.82)
	Dark—lighted or unknown light	2,713 (21.61)
Light condition	Dark—not lighted	2,927 (23.31)
	Dawn	259 (2.06)
	Dusk	276 (2.2)
	1	11,443 (91.14)
Fatalities in crash	2	890 (7.09)
	>2	222 (1.77)
	No	10,543 (83.97)
Speeding	Yes, exceeded speed limit	1,101 (8.77)
	Yes, too fast or racing	911 (7.26)
Traffic control device	No control	9,926 (79.06)
	Traffic control signal (on colors) with pedestrian signal	296 (2.36)
	Traffic control signal (on colors) without or unknown pedestrian signal	1,373 (10.94)
	Stop sign	625 (4.98)
	Warning sign	156 (1.24)
	Other regulatory sign	68 (0.54)
	Flashing traffic control signal	40 (0.32)

Table 3: Frequency and Distribution of Variables Related to Safety, and Crash Related Factors

Variable	Category	Frequency (%)
	Highway traffic signal	9 (0.07)
	Railway crossing device or school zone device	26 (0.21)
	Yield sign or person	36 (0.29)
Presence of curve	Straight	10,352 (82.45)
	Curve	2,109 (16.8)
	Non-trafficway or driveway access	94 (0.75)
Surface condition	Dry	10,774 (85.81)
	Wet	1,435 (11.43)
	Ice/frost	92 (0.73)
	Snow	86 (0.68)
	Mud, dirt, or gravel	40 (0.32)
	Oil or water	34 (0.27)
	Non-Trafficway or Driveway Access	94 (0.75)

### 5. Synthesizing the Safety Effects of Level 1 and Level 2 CAVs

As all the variables were categorical variables, logistic regression techniques, which are most suitable for analysis of categorical variables were explored to identify the potential analysis method.

#### 5.1 Proportional Odds Model

The dependent variable is ordinal, and the aim of this research is to model and identify the risk factors influencing fatal crashes involving level 1 and level 2 CAVs compared to level 0 vehicles. Ordered probability models (ordered probit or logit model), a class of logistic models, are regression models which can be used when the dependent variable has three or more categories, and the order of different categories is important.<sup>57</sup> In this research, the fatal crashes involving vehicles with different levels of automation are analyzed. They follow a hierarchical order as various smart features in level 1 and level 2 CAVs make them safer.<sup>38</sup> Thus, ordered probability models were identified as more appropriate method for the present analysis.

The proportional odds (PO) modeling technique may also be appropriate for analysis of this problem as there are many parameters related to a fatal crash, and all of them may directly or indirectly influence the occurrence of a crash. The PO model provides the odds (likelihood) of occurrence of a particular event for a selected category compared to the base category (optimum condition). One of the fundamental assumptions of the PO model is that the independent variables influence all the categories of the dependent variable identically (equal slope). In other words, for a dependent variable, with order Y= 1, 2,...,p, where p>1, the PO model with 'n' independent variables (X1, X2,...Xn) has (p-1) intercepts with 'n' slopes. The PO model is mathematically expressed as shown in Equation  $1.^{58-61}$ 

$$\ln(Y'_{p}) = logit[\pi(x)] = \ln\left(\frac{\pi(x)}{1-\pi(x)}\right) = \propto_{p} + (\beta_{1}X_{1} + \beta_{2}X_{2} + \beta_{3}X_{3} + \dots + \beta_{n}X_{n})$$
(1)

The prediction of the PO model represents the expected logit for the category 'p' or above and Y'p represents the odds of being in higher categories. In order to estimate the probability, PO model predictions are required to be transformed as odds, which can be used to estimate probability using Equation 2.<sup>61</sup>

$$P(Y \ge p) = \frac{\exp\left(\ln(Y'_p)\right)}{1 + \exp\left(\ln(Y'_p)\right)} \tag{2}$$

The categories of all the independent variables as well as dependent variables along with the frequency and percentages are obtained using Statistical Analysis System (SAS) analytics suite.<sup>57</sup>

#### 5.2 Partial Proportional Odds (PPO) Model

Three fatal crashes involving level 0 vehicles, which were closest in locations to each fatal crash involving level 1 or level 2 CAV, were considered for analysis. Thus, it is assumed that the spatial heterogeneity associated with fatal crashes involving level 0 vehicles would also be associated with fatal crashes involving level 1 or level 2 CAV. Further, the severity level is the same amongst samples as only fatal crashes are considered for comparison. Thus, the partial proportional odds (PPO) model, which allows flexibility by providing unequal and equal slopes to different independent variables, was used instead of other statistical methods which account for heterogeneity.

Prior to developing the PPO model, an odds proportionality test was conducted using SAS. The test showed that all the independent variables are not influencing all the categories of the dependent variable (fatal crash involvements by the level of automation) identically. In other words, some of the factors affecting fatal crash occurrence are different in crashes involving level 1 and level 2 CAVs. The null hypothesis that all the independent variables have an identical effect on all dependent variables is rejected (p-value less than 0.05). Several independent variables have unequal slopes, which violates the basic assumption of equal slopes in the PO model. Thus, the PPO model was developed using SAS by identifying and assigning both equal and unequal slopes to the independent variables, as shown in Equation 3.

$$\ln \left( Y'_{p} \right) = \ln \left( \frac{\pi_{p}(x)}{1 - \pi_{p}(x)} \right) = \propto_{p} + \left( \beta_{1p} X_{1} + \beta_{2p} X_{2} + \beta_{3p} X_{3} + \dots + \beta_{np} X_{n} \right)$$
(3)

#### 5.3 PPO Model Results

The results of the PPO model indicate that all the variables shown in Table 4 are significant, except the number of lanes, at a 90% confidence level. Further, the individual significance of each category of the independent variable 'number of lanes' indicated that the category (road with one lane) is significant at a 99% confidence level, thus the parameter is not dropped from the model. The final model is developed considering all the variables mentioned in Table 4. It is the result of modeling using a backward elimination approach, which was considered to remove one variable at a time which were not significant to improve the model fit.

Independent variable	Wald chi-square	p-value
Vehicles involved	46.18	<.01
Pedestrian involved	7.81	0.01
Bicyclist involved	3.94	0.05
Day of the week	23.34	<.01
Time of day	17.38	0.02
Functional class	23.89	0.02
Manner of collision	138.21	<.01
Portion of the road on which crash occurred	19.57	0.01
Type of intersection	4.69	0.10
Fatalities in crash	12.01	<.01
Vehicle at fault	9.61	<.01
Rollover	15.34	<.01
Damage to the vehicle	15.37	0.02
Speeding	5.91	0.05
Traffic way type	82.45	<.01
Number of lanes	10.08	0.12
Traffic control device	15.02	0.09
Pre-crash stability of vehicle	23.89	<.01
Pre-crash movement of vehicle	10.41	0.06

#### Table 4. Analysis of Effects

The effects of the independent variables (Table 4) also indicates that number of vehicles involved, manner of collision, traffic way type, and pre-crash stability of the vehicle are the most significant independent variables. The results of the PPO model developed using SAS along with the odds ratio are summarized in Table 5. The Akaike information criterion (AIC) value of the PPO model shown in Table 5 is 14978. It was 15496 for the intercept-only model.

Further, the likelihood ratio and Wald chi-square statistic values are also statistically significant (p-value less than 0.01), which indicates that the model with independent variables is a better fit than the intercept-only model. The estimated results in Table 5 also show that three of the independent variables have varying slopes and their influence in fatal crashes involving level 1 and level 2 CAVs varies.

Variable (Reference	Catagory	Estimate Odd		Odds ratio	odds ratio	
category)	Category	Level 1	Level 2	Level 1	Level 2	
Intercept		-1.822*	-2.645*	-	-	
	2	0.086		1.09		
Vehicles involved (1)	3	0.336*		1.4*		
	>=4	0.631*		1.879*		
Pedestrian involved (no)	Yes	0.283*		1.328*		
Bicyclist involved (no)	Yes	0.354**		1.425**		
	Monday	-0.011		0.989		
	Tuesday	-0.112		0.894		
Day of the week	Thursday	-0.112		0.894		
(Wednesday)	Friday	-0.067		0.935		
	Saturday	-0.027		0.974		
	Sunday	0.235**		1.265**		
	0:00–2:59am	-0.302**		0.739*		
	3:00–5:59am	-0.176		0.838		
	6:00–8:59am	-0.182		0.833		
1 me of the day $(9.00-11.59.00 \text{ am})$	12:00–2:59pm	-0.084		0.919		
(7.00 11.57.00aiii)	3:00–5:59pm	0.002		1.002		
	6:00–8:59pm	-0.004		0.996		
	9:00–11:59pm	-0.187**		0.829**		
Functional class (principal arterial— other)	Interstate	-0.026	-0.137	0.974	0.872	
	Principal arterial—other freeways and expressways	0.127	0.238**	1.136	1.269**	
	Minor arterial	0.06	0.055	1.062	1.057	
	Major collector	0.064	0.037	1.066	1.038	
	Minor collector	-0.01	0.251	0.99	1.286	
	Local	0.093	0.105	1.097	1.111	
Manner of collision (angle)	Front-to-front	0.227**		1.257**		
	Front-to-rear	-0.057		0.944		
	Rear-to-side	-0.669		0.512		
	Sideswipe—opposite direction	0.072		1.074		
	Sideswipe—same direction	0.074		1.077		
	The first harmful event was not a collision with a motor vehicle in transport	0.639**		1.894**		

### Table 5: Estimates and Odds Ratios for Models by the Level of Automation

Variable (Reference	Catagoria	Estimate Odds ratio				
category)	Category	Level 1	Level 2	Level 1	Level 2	
	Not a collision with motor vehicle in-transport	-0.244*		0.783*		
	Acceleration/deceleration lane	0.498		1.646		
	Crossover-related	-0.328		0.72		
Portion of the road	Driveway access related	-0.092		0.912		
on which crash	Entrance/exit ramp related	-0.422		0.656		
occurred (non-	Intersection-related	0.625		1.868		
junction)	Other location within interchange area	0.589**		1.803**		
	Railway grade crossing	-0.535		0.586		
	Through road	0.494*		1.639*		
	Three-way intersection	-0.678		0.508		
Type of intersection	Four-way intersection	-0.811**		0.445**		
(not an intersection)	Roundabout, traffic circle, or multiple-way intersection	0		-		
$\mathbf{F}_{1}$	2	0.283*	0.283* 1.327		.327*	
Fatalities in crash (1)	>2	0.184		1.202		
Vehicle at fault (no)	Yes	-0.191**		0.826**		
Rollover (no rollover)	Yes	-0.325**	-0.228**	0.723**	0.796**	
Damage to the vehicle (no damage)	Minor damage	0.132	0.521**	1.141	1.685**	
	Functional damage	0.357*	0.611*	1.43*	1.843*	
	Disabling damage	0.258	0.53**	1.294	1.699**	
Speeding (pa)	Yes, exceeded speed limit	s, exceeded speed limit 0.166		1.181		
Speeding (110)	Yes, too fast or racing	-0.136		0.872		
	One-way trafficway	0.002 1.002		1.002	1.002	
	Two-way, not divided	0.14		1.151		
Traffic way type (two-way divided with positive median barrier)	Two-way, not divided with a continuous left-turn lane	0.173		1.189		
	Two-way, divided, unprotected (painted > 4 feet) median	-2.395*		0.091*		
	Two-way, divided, unprotected median	0.404*		1.499*		
	Entrance/exit ramp	0.181		1.198		
	Non-trafficway or driveway access	0.394		1.482		
Number of lanes (two lanes)	Non-trafficway or driveway access	0		-		

Variable (Reference		Estimate		Odds ratio	
category)	Category	Level 1	Level 2	Level 1	Level 2
	One lane	0.723**		2.068**	
	Three lanes	0.057		1.059	
	Four lanes	0.014		1.015	
	Five lanes	0.011		1.011	
	Six lanes	-0.038		0.963	
	Seven or more lanes	0.001		1.001	
	Traffic control signal (on colors) with pedestrian signal	0.018		1.019	
	Traffic control signal (on colors) without or unknown pedestrian signal	0.089		1.093	
	Stop sign	0.319*		1.375*	
Traffic control device	Warning sign	-0.236		0.79	
(no control)	Other regulatory sign	-0.467		0.627	
	Flashing traffic control signal	0.223		1.249	
	Highway traffic signal	-1.334		0.263	
	Railway crossing device or school zone device	-0.112		0.894	
	Yield sign or person	0.689**		1.991**	
Pre-crash stability of vehicle (tracking)	Skidding laterally	-0.417*		0.659*	
	Skidding longitudinally	-0.443*		0.642*	
	Other vehicle loss of control	-9.692		<0.001	
	Pre-crash stability not specific	0.147**		1.159**	
Pre-crash movement of vehicle (stayed in original travel lane)	Stayed on road (left original travel lane)	-0.13		0.878	
	Departed road	-0.143		0.867	
	Remained off road	-0.348		0.706	
	Entered road	-0.6		0.549	
	Returned to road	-0.845**		0.43**	

Note 1: \*\* Significant at a 95% confidence level Note 2: \* Significant at a 90% confidence level

The negative value of estimates, e.g., for the portion of the road on which a crash occurred (crossover related) and speeding (too fast), indicate a lower likelihood of a particular outcome. The odds ratio for each parameter represents the odds of a specific outcome compared to the other outcomes. It was computed using the exponential of the estimate for a particular category. For

example, for four-way intersections with an estimate (-0.81), the odds ratio is 0.0445 (e-0.081), indicating that a fatal crash involving a level 1 or level 2 CAV is less likely to occur at a four-way intersection compared to a fatal crash involving level 0 vehicle, than at any location on the route other than an intersection. Likewise, the odds of being involved in a fatal crash for level 1 or level 2 CAVs compared with level 0 vehicles are 1.64 times higher on through lanes than at non-junction locations, when keeping all other parameters constants.

Crashes were statistically significantly less likely to occur in two time of the day periods (9:00 pm to 11:59 pm and (0:00 am to 2:59 am) at a 95% confidence level compared with the reference time. Further, it is 27% more likely that fatal crashes involving CAVs occur on Sunday (weekend) than on Wednesday (weekday).

Pedestrians are the most vulnerable road users. The odds of a crash involving pedestrians or bicyclists was greater than 1, indicating a higher likelihood of fatal crashes involving CAVs and pedestrians or bicyclists. Further, from a safety perspective, head-on collisions are generally more severe than angle collisions and the likelihood of a head-on collision is higher compared to an angle collision. Among all the portions on the road, odds that fatal crashes will occur on a through road or at a location within the interchange area are higher.

CAVs are considered safer vehicles, and the model results in the case of rollovers also convey that the likelihood of level 1 and level 2 CAVs being involved in a rollover is less compared to no rollover. In addition, the odds of CAV crash involving minor, functional, and disabling damage is also high compared to no damage, which is as expected in case of a fatal crashes.

Road geometry plays an important role in the occurrence of a crash. The model results show that the likelihood of a fatal crash on a two-way divided road with a median wider than 4 feet is less compared to a two-way road with a positive median barrier. Contrarily, a two-way divided road with a positive median barrier is safer than the road with an unprotected median. Further, the likelihood of a fatal crash on a one-lane road is 107% higher compared to two-lane roads, and the odds of a fatal crash at a stop sign and yield sign are, respectively, 37% and 99% higher than at intersections with no traffic control.

The CAVs with smart features could also influence the driving behavior of drivers, as while relying on technology they could become less attentive, which may be why drivers are less attentive at locations with stop or yield signs. Finally, the likelihood of getting involved in a fatal crash while returning on the road is less compared to the vehicles traveling in the same lane.

The variables such as rollover, damage to the vehicle, and functional class with unequal slopes show that the response of level 1 and level 2 CAVs is not the same. Independent variables such as rollover and damage to the vehicle show that level 1 CAVs are safer than level 2 CAVs, whereas the odds ratio of different functional classes shows that level 2 CAVs are safer than level 1 CAVs on interstates, minor arterials, and major collectors. On principal arterials, minor collectors, and local roads, level 1 CAVs were found to be safer than level 2 CAVs. As level 2 CAVs have both LKA and ACC, they can provide enhanced safety in the case of freeways by controlling the acceleration as well as lane departure, which are often causes of crashes on freeways. Further, the odds of crashing for vehicles traveling too fast are less compared to vehicles that are not overspeeding in the case of CAVs, indicating that CAVs could provide better safety at high speeds, primarily due to the smart features.

### 6. Analysis of Effect of Smart Features on Fatal Crash Occurrence

The involvement of level 1 and/or level 2 CAVs in fatal crashes compared to level 0 vehicles varies with the smart features. Thus, based on the VIN, details regarding fatal crashes involving vehicles with smart features that are designed for performing specific tasks and enhancing safety were extracted. The ACC system controls acceleration and deceleration of vehicle to maintain a significant gap from the vehicle in front,<sup>39</sup> which affects the likelihood of the vehicle being involved in a rear-end collision. If the leading vehicle slows down, the sensor detects that movement and automatically applies brakes. If no vehicles are present in front, the vehicle travels at a set speed.<sup>37</sup> The LKA pushes the vehicle towards center of the lane if it is moving outside the lane and makes the driving task smoother. The PAEB system automatically engages brakes when it identifies a pedestrian in front of the vehicle and reduces chances of a crash with the pedestrian.<sup>37</sup> The blind spot monitoring (BSM) system, also known as side-view assist system, provides a warning to the driver when other vehicles or objects are present in the blind spot of the vehicle and improves the safety from sideswipe or rear-to-side collisions.<sup>38</sup> The FCWS provides a warning to the vehicle in the case of forward collision risk and reduces the chances of being involved in rear-end collisions when the driver is at fault.<sup>38</sup> The data for all the features and different collision types for which the specific features are designed for was extracted from the FARS database and summarized as shown in Table 6. Further, similar to the previous models, data for years 2016 to 2019 is used for the purpose of comparison.

In Table 6, the proportion of pedestrian crashes from all fatal crashes is higher than the corresponding proportion for vehicles equipped with a PAEB system. In the case of ACC and FCWS, the percentage of rear-end collisions is almost double compared to other vehicles without these features. Similarly, the proportion of sideswipe or rear-to-side collision of vehicles with BSM is also double compared to the data of all crashes. In the case of vehicles with LKA system, fatal crashes involving roadside departure are considered for the analysis. Crashes involving roadside departure include only cases when the vehicle runs off the road. Other cases where the vehicle ran off the road while trying to avoid collision with other vehicles/pedestrians or due to traction loss are not considered in this comparison. These trends can be mainly attributed to only considering fatal crash data in the evaluation. Considering injury crashes and property damage crashes in the evaluation could further improve the clarity of these findings.

Overall, smart features are designed to enhance safety and avoid particular types of crashes. However, research findings indicate that these smart features may not be yet effective all the time. Potential reasons are related to localization and the ability of features to sense and control the CAV irrespective of the geographic location and geometric, environmental, and traffic conditions.

Safety features	Total crashes	Pedestrian crashes	Rear-end collisions	Rear-to-side collisions or sideswipe	Total vehicles involved	Roadside departure
# of Crashes	1,36,471	26,366 (19.32%)	9,650 (7.07%)	324 (1.70%)	2,09,375	30,374 (14.51%)
Crashes involving vehicles with PAEB system	1,327	217 (16.35%)	_	-	_	_
Crashes involving vehicles with FCWS	2,889	-	425 (14.71%)	-	-	_
Crashes involving vehicles with ACC	2,356	-	346 (14.69%)	_	_	_
Crashes involving vehicles with BSM	2,253	-	-	77 (3.42%)	-	-
Vehicles with LKA	-	-	-	-	2,845	288 (10.12%)

Table 6: Comparison of Vehicles with Smart Features and All Vehicles

### 7. Summary & Conclusions

#### 7.1 Summary

The risk factors influencing fatal crashes involving level 1 and level 2 CAVs compared to level 0 vehicles within vicinity were explored using PO and PPO models. The PO test showed that all the independent variables do not have equal slopes, and their effect on the dependent variable varies, which violated the basic assumption of the PO model. Thus, the PPO model was considered appropriate and developed using all the independent variables.

The factors such as rollover of vehicles, pre-crash movement of a vehicle, type of intersection, and pre-crash stability of a vehicle indicate that the likelihood of a fatal crash for level 1 and level 2 CAVs is less compared to level 0 vehicles. The CAVs are safer when vehicles depart the travel lane or skid laterally or longitudinally before the crash, indicating their safety benefits in snow or rainy weather. Level 1 and level 2 CAVs are safer on interstates and in conditions when the vehicle is moving too fast than the speed limit, ensuring their higher safety, especially on high-speed routes. Further, the odds of a fatal crash involving level 2 CAVs are safer on low-volume roads such as minor collectors and local roads. Further, the chances of a fatal crash involving CAVs on a two-lane road is lower than a road with single or multiple lanes.

The chances of level 1 or level 2 CAVs getting involved in a fatal crash with non-motorists such as pedestrians or bicyclists is higher than level 0 vehicles. One of the reasons behind the higher odds of getting involved in fatal crashes with non-motorists may be due to decreased attentiveness of drivers due to reliance on the technology. Similarly, the chances of a fatal crash involving a CAV at a flashing signal or stop sign is also higher than a level 0 vehicle. The odds of more than one fatality in crashes involving level 1 or level 2 CAVs are also higher compared to level 0 vehicles. CAVs are more likely to be involved in fatal crashes on different portions of the road, like in acceleration or deceleration lanes, intersections, and any location within interchange areas.

LKA usually works based on the information regarding lane markings provided by the cameras attached to the vehicle. Locations like intersections with multiple or no markings may increase the uncertainty of such systems when making decisions, which could be the potential reason for the high likelihood of fatal crashes at those locations. The likelihood of a rear-end collision or rear-to-side collision is less compared to a head-on collision or sideswipe collision.

The odds ratio of different variables such as functional class, traffic control devices, traffic way type, the portion of the road, and manner of collision vary significantly for level 1 and level 2 CAVs compared to level 0 vehicles. Therefore, an analysis to identify the effect of smart safety features was carried out. The findings indicate that the vehicles with smart features are still involved in specific types of crashes for which they are designed to avoid. Overall, parameters such as pre-

crash stability and movement of a vehicle, and crash at high speed, which are primarily related to the vehicles, indicate that level 1 and level 2 CAVs are safer. In contrast, fatal crash involvement with non-motorists and at stop and yield signs where drivers have to make decisions (be attentive) had a higher likelihood of occurrence in level 1 and level 2 CAVs. Therefore, level 1 and level 2 CAVs are improving the overall safety of vehicles. At the same time, it also affects drivers' attentiveness, resulting in an increased likelihood of a fatal crash in some situations.

#### 7.2 Conclusions

This research aimed to identify the effect of level 1 and level 2 CAVs on the occurrence of fatal crashes. A review of existing studies on the penetration of level 1 and level 2 CAVs, the effect of CAVs on mode choice and travel behavior, factors influencing fatal crashes, and the occurrence of fatal crashes was carried out. Fatal crash data obtained from the FARS database and smart feature data based on VINs retrieved from the NHTSA database were combined to identify the level 1 and level 2 CAVs. To identify the effect of levels of automation on fatal crash occurrence, a PPO model was developed comparing level 0 vehicles with level 1 and level 2 CAVs involved in fatal crashes. A comparative analysis of vehicles equipped with smart features and other vehicles was carried out to determine the efficiency of smart features designed to enhance safety in various types of crashes.

A nearest neighborhood analysis was carried out to identify crashes involving level 0 vehicles that occurred in the vicinity of crashes involving level 1 and level 2 CAVs. The PPO model was developed based on the identified sample of crashes involving level 1, level 2 CAVs, and three nearest crashes involving level 0 vehicles.

The results of the PPO model indicate that the involvement of level 1 and level 2 CAVs in fatal crashes with pedestrians and bicyclists is higher compared to level 0 vehicles which may be due to the inattentiveness or overreliance of drivers on smart features. Further, level 1 and level 2 CAVs have higher odds of getting involved in fatal crashes on one-lane routes and near locations with stop, yield, or flashing yellow signs. The CAVs were also found not to be safe in the case of head-on collision and collision with non-motorists.

The vehicles with smart features such as ACC and FCWS, which are designed to improve safety in the case of rear-end collisions, are getting involved in such collisions more than level 0 vehicles. However, LKA and PAEB systems are more efficient than level 0 vehicles in the case of roadside departures and crashes with pedestrians.

The findings of this research provide an overview of the factors influencing the occurrence of fatal crashes involving level 1 and level 2 CAVs. These research findings can assist automobile manufacturers in modifying the existing technologies underlying various smart features engaged in level 1 and level 2 CAVs. In addition, the overview provided by this research can aid

practitioners and transportation engineers to better plan, design facilities, and modify existing policies.

#### 7.3 Limitations and Future Scope

The primary purpose of this research was to identify the effect of smart features on safety. The safety performance of a vehicle may vary depending on the manufacturer of the vehicle and the type or functionality of the technology. For example, ACC works either based on a camera or radar. The effect of different technologies on which ACC works was not identified or explored in this research. Thus, an analysis incorporating the manufacturer and the type of technology as independent variables would provide insights on the variation in the safety performance of different vehicles with the same level of automation. Further, the PPO model used in this research is a fixed parameters model, which restricts parameters to remain the same for all observations. However, random parameters modeling allows each observation to have its own parameter estimate and can account for variability in individual crash-specific characteristics.<sup>55</sup> Such an approach would be more applicable as larger and more detailed datasets (with the type of technology and their functionality at the time of crash by vehicle manufacturer) become available.

The proportion of level 1 and level 2 CAVs is very low, as only 1.8% of all fatal crashes in the years studied involved either a level 1 or level 2 CAV. Therefore, most of the crashes involving level 1 and level 2 CAVs are with level 0 vehicles. Thus, analysis at higher penetration rates may yield different results and merits further investigation. In addition, the analysis considering only at fault vehicles would also provide insights about the risk driver possess to other drivers while driving vehicles with different levels of automation.

The effect of levels of automation was identified using the crash data covering four years in this research. However, analysis to accommodate the variation in risk factors influencing crashes with time may yield better results.

## Appendix A

This appendix summarizes the make of level 1 and level 2 CAVs identified based on VIN from the FARS database.

Make	Level 1 CAVs	Level 2 CAVs
Acura	2	21
Alfa Romeo	1	0
Audi	0	21
BMW	44	35
Buick / Opel	6	7
Cadillac	46	23
Chevrolet	305	19
Chrysler	16	27
Dodge	40	65
Ford	123	229
GMC	56	38
Honda	33	316
Hyundai	29	113
Infiniti	0	18
Jaguar	3	0
Jeep / Kaiser-Jeep / Willys- Jeep	72	159
KIA	145	14
Land Rover	1	17
Lexus	10	33
Lincoln	4	20
Mazda	3	43
Mercedes-Benz	2	77
Nissan/Datsun	3	6
Other Domestic Manufacturers	0	28
Other Import	6	5
Porsche	3	8
Subaru	20	123
Toyota	190	529
Volkswagen	4	38
Volvo	1	16

Table A1. Vehicle Make Details of Level 1 and Level 2 CAVs Involved in Fatal Crashes

# Abbreviations and Acronyms

AADT	Annual Average Daily Traffic
ABS	Automated Braking System
ACC	Adaptive Cruise Control
AIC	Akaike Information Criterion
BSM	Blind Spot Monitoring
CAV	Connected and Automated Vehicle
CDC	Centers for Disease Control and Prevention
FARS	Fatality Analysis Reporting System
FCWS	Forward Collision Warning System
LKA	Lane-Keeping Assistance
NHTSA	National Highway Traffic Safety Administration
PAEB	Pedestrian Automatic Emergency Braking
РО	Proportional Odds
PPO	Partial Proportional Odds
SAE	Society of Automotive Engineers
SAV	Shared Automated Vehicles
VIN	Vehicle Identification Number

### Endnotes

- 1. CDC (Center of Disease Control and Prevention), *The top 10 causes of death, 2020*, accessed March 12, 2021, https://webappa.cdc.gov/cgi-bin/broker.exe.
- 2. NHTSA (National Highway Traffic Safety Administration), *Traffic safety facts*, 2019, accessed June 28, 2021, https://crashstats.nhtsa.dot.gov/Api/Public/ViewPublication/813060.
- 3. NSC, *Guide to calculating cost, injury facts. 2019*, accessed April 18, 2021, https://injuryfacts.nsc.org/all-injuries/costs/guide-to-calculating-costs/data-details.
- 4. CDC (Center of Disease Control and Prevention), Cost data and prevention policies. Web-based Injury Statistics Query and Reporting System (WISQARS), 2018.
- 5. NHTSA (National Highway traffic Safety Administration), *Traffic safety facts: research notes: distracted driving 2015, 2017,* accessed September 28, 2021, https://crashstats.nhtsa.dot.gov/api/public/viewpublication/812381.
- 6. J.C.F. de Winter, et al., "Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence," *Transportation Research Part F: Traffic Psychology and Behaviour* 27 (2014): 196-217.
- 7. A. Smiley, "Behavioral Adaptation, Safety, and Intelligent Transportation Systems," *Transportation Research Record* 1724 no.1 (2000): 47-51.
- 8. J. Son, M. Park, and B.B. Park, "The effect of age, gender and roadway environment on the acceptance and effectiveness of Advanced Driver Assistance Systems," *Transportation Research Part F-traffic Psychology and Behaviour* 31 (2015): 12-24.
- 9. A. Mehrara Molan, M. Moomen, and K. Ksaibati, "Investigating the effect of geometric dimensions of median traffic barriers on crashes: Crash analysis of interstate roads in Wyoming using actual crash datasets." *Journal of Safety Research* 71 (2019): 163-171.
- A. M. Molan, M. Moomen, and K. Ksaibati, "Estimating the effect of geometric features of side traffic barriers on crash severity of interstate roads in Wyoming." *Accident Analysis & Prevention* 144 (2020): 105639.
- 11. A.C. Wagenaar, et al., "Effects of legal BAC limits on fatal crash involvement: analyses of 28 states from 1976 through 2002." *Journal of Safety Research* 38 no.5 (2007): 493-499.
- 12. R.B. Noland, and L. Oh, "The effect of infrastructure and demographic change on traffic-related fatalities and crashes: a case study of Illinois county-level data," *Accident Analysis & Prevention* 36 no.4 (2004): 525-532.
- 13. S. Chen, et al., "Safety sensitivity to roadway characteristics: A comparison across highway classes," *Accident Analysis & Prevention* 123 (2019): 39-50.
- 14. P.A. Pisano, "U.S. highway crashes in adverse road weather conditions," in 24th Conference on IIPS. 2008.

- 15. S. Saha, et al., "Adverse weather conditions and fatal motor vehicle crashes in the United States, 1994-2012." *Environmental health: a global access science source* 15 no.1 (2016): 104-104.
- 16. S.S. Pulugurtha, and R. Otturu, "Effectiveness of red light running camera enforcement program in reducing crashes: Evaluation using "before the installation", "after the installation", and "after the termination" data," *Accident Analysis & Prevention* 64 (2014): 9-17.
- 17. W. Hu, and J.B. Cicchino, "Effects of turning on and off red light cameras on fatal crashes in large U.S. cities," *Journal of Safety Research* 61 (2017): 141-148.
- 18. R. Tay, "A random parameters probit model of urban and rural intersection crashes." *Accident Analysis* & *Prevention* 84 (2015): 38-40.
- 19. S. Chen, T.U. Saeed, and S. Labi, "Impact of road-surface condition on rural highway safety: A multivariate random parameters negative binomial approach," *Analytic Methods in Accident Research*, 16 (2017): 75-89.
- 20. S.S. Pulugurtha, and A.P. Nujjetty, "Assessment of models to estimate crashes at intersections: with and without using traffic volume," in 91st Annual Meeting of Transportation Research Board. 2012, Transportation Research Board: Washington D. C.
- 21. C. M. Farmer, et al., "Fatal crashes of passenger vehicles before and after adding antilock braking systems," *Accident Analysis & Prevention* 29 no.6 (1997): 745-757.
- 22. C.S. Crandall, L.M. Olson, and D.P. Sklar, "Mortality reduction with air bag and seat belt use in head-on passenger car collisions," *American Journal of Epidemiology* 153 no.3 (2001): 219-224.
- P. Cummings, "Association of seat belt use with death: a comparison of estimates based on data from police and estimates based on data from trained crash investigators," *Injury Prevention* 8 no.4 (2002): 338.
- 24. A.F. Williams, S.A. Ferguson, and J.K. Wells, "Sixteen-year-old drivers in fatal crashes, United States, 2003," *Traffic Injury Prevention* 6 no.3 (2005): 202-206.
- 25. N. Gruber, et al., "Vision and night driving abilities of elderly drivers," *Traffic Injury Prevention* 14 no.5 (2013): 477-485.
- A. Meng, and A. Siren, "Cognitive problems, self-rated changes in driving skills, driving-related discomfort and self-regulation of driving in old drivers," *Accident Analysis & Prevention* 49 (2012): 322-329.
- 27. SAE (Society of Automotive Engineers), International, taxonomy and definitions for terms related to driving automation systems for on-road motor vehicles. 2018.
- 28. Jakob. *The 6 levels of autonomous driving, open autonomous driving,* 2018, accessed July 12, 2021, https://autonomous-driving.org/2018/03/20/the-6-levels-of-autonomous-driving/.
- 29. J.S. Choksey, and C. Wardlaw. *Levels of Autonomous Driving Explained*, 2021, accessed July 28, 2021, https://www.jdpower.com/cars/shopping-guides/levels-of-autonomous-driving-explained.

- 30. M. Makridis, et al., "Assessing the impact of connected and automated vehicles," *A freeway scenario* (2018): 213-225.
- J. Shelton, et al., "Impacts of connected vehicles in a complex, congested urban freeway setting using multi-resolution modeling methods," *International Journal of Transportation Science and Technology*, 8 no.1 (2019): 25-34.
- 32. H. Abdulsattar, et al., "Measuring the impacts of connected vehicles on travel time reliability in a work zone environment: an agent-based approach," *Journal of Intelligent Transportation Systems* 24 no.5 (2020): 421-436.
- 33. J. Rios-Torres, A. Malikopoulos, and P. Pisu, Online optimal control of connected vehicles for efficient traffic flow at merging roads, 2015: 2432-2437.
- 34. S. Mathew, S. Duvvuri, and S. Pulugurtha, Modeling and assessing the influence of connected and automated vehicle penetration rates on urban arterial road operational performance. 2020: 98-108.
- 35. A. Deluka Tibljaš, et al., "Introduction of autonomous vehicles: roundabouts design and safety performance evaluation," *Sustainability* 10 no.4 (2018).
- 36. N. Virdi, et al., "A safety assessment of mixed fleets with Connected and Autonomous Vehicles using the Surrogate Safety Assessment Module," *Accident Analysis & Prevention* 131 (2019):95-111.
- 37. A.H. Eichelberger, and A.T. McCartt, "Volvo drivers' experiences with advanced crash avoidance and related technologies," *Traffic Injury Prevention* 15 no. 2 (2014): 187-195.
- 38. J.S. Jermakian, "Crash avoidance potential of four passenger vehicle technologies," Accident Analysis & Prevention 43 no.3 (2011): 732-740.
- 39. Y. Li, et al., "Evaluating the safety impact of adaptive cruise control in traffic oscillations on freeways," *Accident Analysis & Prevention* 104 (2017): 137-145.
- S.M. Gaweesh, M.M. Ahmed, and A.V. Piccorelli, "Developing crash prediction models using parametric and nonparametric approaches for rural mountainous freeways: A case study on Wyoming Interstate 80," *Accident Analysis & Prevention* 123 (2019): 176-189.
- 41. J.-K., Kim, Y. Wang, and F. Ulfarsson Gudmundur, "Modeling the probability of freeway rear-end crash occurrence," *Journal of Transportation Engineering* 133 no.1 (2007): 11-19.
- N.N. Sze, S.C. Wong, and C.Y. Lee, "The likelihood of achieving quantified road safety targets: A binary logistic regression model for possible factors," *Accident Analysis & Prevention* 73 (2014): 242-251.
- N. Venkataraman, G.F. Ulfarsson, and V.N. Shankar, "Random parameter models of interstate crash frequencies by severity, number of vehicles involved, collision and location type," *Accident Analysis & Prevention* 59 (2013): 309-318.
- Y. Li, et al., "Evaluating impacts of different longitudinal driver assistance systems on reducing multivehicle rear-end crashes during small-scale inclement weather," *Accident Analysis & Prevention* 107 (2017): 63-76.

- 45. A. Sinha, et al., "Comprehensive safety assessment in mixed fleets with connected and automated vehicles: A crash severity and rate evaluation of conventional vehicles," *Accident Analysis & Prevention* 142 (2020): 105567.
- 46. E.R. Teoh, and D.G. Kidd, "Rage against the machine? Google's self-driving cars versus human drivers," *Journal of Safety Research* 63 (2017): 57-60.
- 47. C. Xu, et al., "Statistical analysis of the patterns and characteristics of connected and autonomous vehicle involved crashes," *Journal of Safety Research* 71 (2019): 41-47.
- 48. A. Sinha, et al., "A crash injury model involving autonomous vehicle: investigating of crash and disengagement reports," *Sustainability* 13 no. 14 (2021).
- 49. M. Lavasani, X. Jin, and Y. Du, "Market penetration model for autonomous vehicles on the basis of earlier technology adoption experience," *Transportation Research Record* 2597 no.1 (2016): 67-74.
- 50. Chen, Z., et al., Optimal deployment of autonomous vehicle lanes with endogenous market penetration. Transportation Research Part C: Emerging Technologies, 2016. 72: p. 143-156.
- 51. L. Kröger, T. Kuhnimhof, and S. Trommer, "Does context matter? A comparative study modelling autonomous vehicle impact on travel behaviour for Germany and the USA," *Transportation Research Part A: Policy and Practice* 122 (2019): 146-161.
- P. Bansal, and K.M. Kockelman, "Forecasting Americans' long-term adoption of connected and autonomous vehicle technologies," *Transportation Research Part A: Policy and Practice* 95 (2017): 49-63.
- 53. J. Nieuwenhuijsen, et al., "Towards a quantitative method to analyze the long-term innovation diffusion of automated vehicles technology using system dynamics," *Transportation Research Part C: Emerging Technologies* 86 (2018): 300-327.
- 54. T. Cover, and P. Hart, "Nearest neighbor pattern classification," *IEEE Transactions on Information Theory* 13 no.1 (1967): 21-27.
- 55. J. Aguero-Valverde, and P.P. Jovanis, "Spatial analysis of fatal and injury crashes in Pennsylvania," *Accident Analysis & Prevention* 38 no.3 (2006): 618-625.
- 56. S. Mitra, "Spatial autocorrelation and Bayesian spatial statistical method for analyzing intersections prone to injury crashes," *Transportation Research Record*, 2136 no.1 (2009): p. 92-100.
- 57. L. Sasidharan, and M. Menéndez, "Partial proportional odds model—An alternate choice for analyzing pedestrian crash injury severities," *Accident Analysis & Prevention* 72 (2014): 330-340.
- 58. P. McCullagh, J.A.N., Generalized Linear Models, Second Edition. (Chapman and Hall/CRC, 1989).
- 59. A.A. O'Connell, "Logistic regression models for ordinal response variables," *Quantitative Applications in the Social Sciences (*Ohio State University, USA: Sage Publication, Inc., 2005)
- 60. N. Eluru, and S. Yasmin, "A note on generalized ordered outcome models," *Analytic Methods in Accident Research* 8 (2015): 1-6.

61. V.R. Duddu, V.M. Kukkapalli, and S.S. Pulugurtha, "Crash risk factors associated with injury severity of teen drivers," *LATSS Research* 2019.

### Bibliography

- Abdulsattar, H., A. Mostafizi, M. R. Siam, and H. Wang. "Measuring the Impacts of Connected Vehicles on Travel Time Reliability in A Work Zone Environment: An Agent- Based Approach." *Journal of Intelligent Transportation Systems* 24 no.5 (2020):421-436.
- Aguero-Valverde, J., and P. P. Jovanis. "Spatial Analysis of Fatal and Injury Crashes in Pennsylvania." Accident Analysis & Prevention 38 no.3 (2006):618-625.
- Bansal, P., and K. M. Kockelman. "Forecasting Americans' Long-term Adoption of Connected and Autonomous Vehicle Technologies." *Transportation Research Part A: Policy and Practice* 95 (2017):49-63.
- Centers for Disease Control and Prevention (CDC). Cost Data and Prevention Policies. Webbased Injury Statistics Query and Reporting System (WISQARS), National Center for Injury Prevention and Control (NCIPC), 2018. Accessed on March 14, 2021. https://www.cdc.gov/injury/wisqars/cost/
- Centers for Disease Control and Prevention (CDC). Leading Cause of Death Reports, 1981-2019. *Web-based Injury Statistics Query and Reporting System (WISQARS)*, National Center for Injury Prevention and Control (NCIPC), 2020. Accessed on March 12, 2021. https://webappa.cdc.gov/cgi-bin/broker.exe
- Chen, S., T. Saeed, and S. Labi. "Impact of Road-Surface Condition on Rural Highway Safety: A Multivariate Random Parameters Negative Binomial Approach." *Analytical Methods in Accident Research* 16 (2017):75-89.
- Chen, S., T. Saeed, M. Alinizzi, S. Lavrenz, and S. Labi. "Safety Sensitivity to Roadway Characteristics: A Comparison across Highway Classes." *Accident Analysis & Prevention* 123 (2019):39-50.
- Chen, Z., F. He, L. Zhang, and Y. Yin. "Optimal Deployment of Autonomous Vehicle Lanes with Endogenous Market Penetration." *Transportation Research Part C: Emerging Technologies* 72 (2016):143-156.
- Choksey J. S., and C. Wardlaw." Levels of Autonomous Driving Explained." J. D. Power Inc. Published on May 05, 2021. Accessed on July 28, 2021. https://www.jdpower.com/cars/shopping-guides/levels-of-autonomous-driving-explained.
- Cover, T., and P. Hart. "Nearest Neighbor Pattern Classification." *IEEE Transactions on Information Theory* 13 no.1 (1967):21-27.

- Crandall C. S., L. M. Olson, and D. P. Sklar. Mortality Reduction with Air Bag and Seat Belt Use in Head-on Passenger Car Collisions. *American Journal of Epidemiology*, 2001. 153(3):219–224.
- Cummings P. "Association of Seat Belt Use with Death: A Comparison of Estimates Based on Data from Police and Estimates Based on Data from Trained Crash Investigators." *Injury Prevention* 8 (2002):338-341.
- De Winter, J. C., R. Happee, M. H. Martens, and N. A. Stanton, "Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence." *Transportation research part F: traffic psychology and behaviour* 27 (2014):196-217.
- Deluka Tibljaš, A., T. Giuffrè, S. Surdonja, and S. Trubia. "Introduction of Autonomous Vehicles: Roundabouts Design and Safety Performance Evaluation." *Sustainability*, 10 (2018):1-14.
- Duddu V. R., V. M. Kukkapalli, and S. S. Pulugurtha. "Crash Risk Factors Associated with Injury Severity of Teen Drivers." *LATSS Research* 43 (2019):37-43.
- Eichelberger, A. H., and A. McCartt. "Volvo Drivers' Experiences with Advanced Crash Avoidance and Related Technologies." *Traffic Injury Prevention*, 15 no.2 (2014):187-195.
- Eluru, N., and S. Yasmin. "A Note on Generalized Ordered Outcome Models." *Analytic Methods in Accident Research* 8 (2015):1-6.
- Farmer, C., A. Lund, R. Trempel, and E. Braver. "Fatal Crashes of Passenger Vehicles Before and After Adding Antilock Braking Systems." *Accident Analysis & Prevention* 29 no.6 (1997):745-757.
- Gaweesh, S. M., M. M. Ahmed, and A. V. Piccorelli. "Developing Crash Prediction Models using Parametric and Nonparametric Approaches for Rural Mountainous Freeways: A Case Study on Wyoming Interstate 80". *Accident Analysis & Prevention* 123 (2019):176-189.
- Gruber, N., U. P. Mosimann, R. M. Müri, and T. Nef. "Vision and Night Driving Abilities of Elderly Drivers." *Traffic Injury Prevention*, 14 no.5 (2013):477-485.
- Hu, W., and J. Cicchino. "Effects of Turning on and off Red-Light Cameras on Fatal Crashes in Large U.S. Cities." *Journal of Safety Research* 61 (2017):141-148.
- Jakob. "The 6 Levels of Autonomous Driving." Open Autonomous Driving. Org, Published on March 20, 2018. Accessed on July 12, 2021. https://autonomous-driving.org/2018/03/20/the-6-levels-of-autonomous-driving/.

- Jermakian, J. S. "Crash Avoidance Potential of Four Passenger Vehicle Technologies." Accident Analysis & Prevention 43 no.3 (2011):732-740.
- Kim, J. K., Y. Wang, and G. F. Ulfarsson. "Modeling the Probability of Freeway Rear-End Crash Occurrence." *Journal of Transportation Engineering* 133 no.1 (2007):11-19.
- Kröger, L., T. Kuhnimhof, and S. Trommer. "Does context matter? A Comparative Study Modelling Autonomous Vehicle Impact on Travel Behaviour for Germany and the USA." *Transportation Research Part A: Policy and Practice* 122 (2019):146-161.
- Lavasani, M., X. Jin, and Y. Du. "Market Penetration Model for Autonomous Vehicles on the Basis of Earlier Technology Adoption Experience." *Transportation Research Record* 2597 no.1 (2016):67-74.
- Li Y., Z. Li, H. Wang, W. Wang, and L. Xing. "Evaluating the Safety Impact of Adaptive Cruise Control in Traffic Oscillations on Freeways." *Accident Analysis & Prevention* 104 (2017a):137-145.
- Li, Y., L. Xing, W. Wang, H. Wang, C. Dong, and S. Liu. "Evaluating Impacts of Different Longitudinal Driver Assistance Systems on Reducing Multi-Vehicle Rear-End Crashes During Small-Scale Inclement Weather." *Accident Analysis & Prevention* 107 (2017b):63-76.
- Makridis, M., K. Mattas, B. Ciuffo, M. A. Raposo, and C. Thiel. "Assessing the Impact of Connected and Automated Vehicles. A Freeway Scenario." In: Zachäus C., Müller B., Meyer G. (eds) Advanced Microsystems for Automotive Applications 2017. Lecture Notes in Mobility. Springer, Cham, 2018, 213-225.
- Mathew, S., S. Duvvuri, and S. S. Pulugurtha. Modeling and "Assessing the Influence of Connected and Automated Vehicle Penetration Rates on Urban Arterial Road Operational Performance." *Proceedings in International Conference on Transportation and Development*, American Society of Civil Engineers, Reston, VA (2020): 98-108.
- McCullagh, P., and J. A. Nelder. *Generalized Linear Models* (Second Edition). Routledge: Boca Raton, 1983.
- Meng A., A. Siren. "Cognitive Problems, Self-Rated Changes in Driving Skills, Driving- Related Discomfort, and Self-Regulation of Driving in Old Drivers." *Accident Analysis & Prevention*, 49 (2012):322-329.
- Mitra, S. "Spatial Autocorrelation and Bayesian Spatial Statistical Method for Analyzing Intersections Prone to Injury Crashes." *Transportation Research Record* 2136 no.1 (2009):92-100.

- Molan, A. M., M. Moomenc, and K. Ksaibati. "Estimating the Effect of Geometric Features of Side Traffic Barriers on Crash Severity of Interstate Roads in Wyoming." *Accident Analysis & Prevention* 144 (2020):105639.
- Molan, A. M., M. Moomenc, and K. Ksaibati. "Investigating the Effect of Geometric Dimensions of Median Traffic Barriers on Crashes: Crash Analysis of Interstate Roads in Wyoming Using Actual Crash Datasets." *Journal of Safety Research* 71 (2019):163-171.
- National Highway Traffic Safety Administration (NHTSA). *Traffic Safety Facts*. 2019. Accessed on June 28, 2021. https://crashstats.nhtsa.dot.gov/Api/Public/ViewPublication/813060
- National Highway Traffic Safety Administration (NHTSA). Traffic Safety Facts: Research Notes:DistractedDriving.2015.Accessedon2021.https://crashstats.nhtsa.dot.gov/api/public/viewpublication/812381
- Nieuwenhuijsen, J., G. H. de Almeida Correia, D. Milakis, B.van Arem, and E. van Daalen, E. "Towards a Quantitative Method to Analyze the Long-term Innovation Diffusion of Automated Vehicles Technology Using System Dynamics." *Transportation Research Part C: Emerging Technologies* 86 (2018):300-327.
- Noland, R., and L. Oh. "The Effect of Infrastructure and Demographic Change on Traffic-Related Fatalities and Crashes: A Case Study of Illinois County-Level Data." *Accident Analysis* & Prevention 36 (2004):525-532.
- NSC, Guide to Calculating Cost. 2019. Guide to Calculating Cost, Injury Facts. 2019. Accessed on April 18, 2021. https://injuryfacts.nsc.org/all-injuries/costs/guide-to-calculating-costs/data-details.
- O'Connell, A. A. Logistic Regression Models for Ordinal Response Variables. Sage Publication, Inc., 2006.
- Pisano P. A., L. C. Goodwin, and M. A. Rossetti. "U.S. Highway Crashes in Adverse Road Weather Conditions." Presented at the 24th Conference on International Interactive Information and Processing Systems for Meteorology, Oceanography and Hydrology, New Orleans, LA, 20-24 January, 2008.
- Pulugurtha, S. S. and A. P. Nujjetty. "Assessment of Models to Estimate Crashes at Intersections: With and Without Using Traffic Volume." In the *Transportation Research Board 91st Annual Meeting, Compendium of Papers, DVD-ROM*, Washington, DC, January 22-26, 2012.

- Pulugurtha, S. S. and R. Otturu. "Effectiveness of Red Light Running Camera Enforcement Program in Reducing Crashes: Evaluation using "Before the Installation", "After the Installation", and "After the Termination" Data." Accident Analysis & Prevention. 64 (2013):9-17.
- Rios-Torres, J., A. Malikopoulos, and P. Pisu. "Online Optimal Control of Connected Vehicles for Efficient Traffic Flow at Merging Roads." *Proceedings in IEEE 18th International Conference on Intelligent Transportation Systems*, Canary Islands, September 2015, 2432-2437.
- Saha, S., P. Schramm, A. Nolan, and J. Hess. "Adverse Weather Conditions and Fatal Motor Vehicle Crashes in the United States, 1994-2012." *Environmental Health* 15 no.104 (2016):1-9.
- Sasidharan, L., and M. Menéndez. "Partial Proportional Odds Model—an Alternate Choice for Analyzing Pedestrian Crash Injury Severities." *Accident Analysis & Prevention* 72 (2014):330-340.
- Shelton, J., J. Wagner, S. Samant, G. Goodin, T. Lomax, and E. Seymour. "Impacts of Connected Vehicles in a Complex, Congested Urban Freeway Setting Using Multi- Resolution Modeling Methods." *International Journal of Transportation Science and Technology* 8 no.1 (2018):25-34.
- Sinha, A., S. Chand, K. P. Wijayaratna, N. Virdi, and V. Dixit. "Comprehensive Safety Assessment in Mixed Fleets with Connected and Automated Vehicles: A Crash Severity and Rate Evaluation of Conventional Vehicles." *Accident Analysis & Prevention* 142(2020): 105567.
- Sinha, A., V. Vu, S. Chand, K. Wijayaratna, and V. Dixit. A Crash "Injury Model Involving Autonomous Vehicle: Investigating of Crash and Disengagement Reports." *Sustainability* 13 no.14 (2021):7938.
- Smiley, A. Behavioral adaptation, safety, and intelligent transportation systems. *Transportation Research Record*, 2000. 1724(1):47-51.
- Society of Automotive Engineers (SAE). International, Taxonomy and Definitions for Terms Related to Driving Automation Systems for On-Road Motor Vehicles. The Society of Automotive Engineers, Article No. J3016\_201806 (2018).
- Son, J., M. Park, and B. B. Park. The effect of age, gender and roadway environment on the acceptance and effectiveness of Advanced Driver Assistance Systems. *Transportation research part F: traffic psychology and behaviour* 31 (2015), 12-24.

- Sze, N. N., S. C. Wong, and C. Y. Lee. "The likelihood of achieving quantified road safety targets: A binary logistic regression model for possible factors." *Accident Analysis & Prevention* 73 (2014):242-251.
- Tay, R. "A Random Parameters Probit Model of Urban and Rural Intersection Crashes." Accident Analysis & Prevention 84 (2015):38-40.
- Teoh, E. R., and D. G. Kidd. "Rage Against the Machine? Google's Self-Driving Cars Versus Human Drivers." *Journal of Safety Research* 63 (2017):57-60.
- Venkataraman, N., G. F. Ulfarsson, and V. N. Shankar. "Random Parameter Models of Interstate Crash Frequencies by Severity, Number of Vehicles Involved, Collision and Location Type." *Accident Analysis & Prevention* 59 (2013):309-318.
- Virdi, N., H. Grzybowska, S. Waller, and V. Dixit. "A Safety Assessment of Mixed Fleets with Connected and Autonomous Vehicles using The Surrogate Safety Assessment Module." *Accident Analysis & Prevention* 131 (2019.):95-111.
- Wagenaar, A. C., M. Maldonado-Molina, L. Ma, A. Tobler, and K. Komro. "Effects of Legal BAC Limits on Fatal Crash Involvement: Analyses of 28 States from 1976 through 2002." *Journal of Safety Research* 38 no.5 (2007):493-499.
- Williams, A. F., S. A. Ferguson, and J. K. Wells. "Sixteen-Year-Old Drivers in Fatal Crashes, United States, 2003." *Traffic Injury Prevention* 6 no.3 (2005):202-206.
- Xu, C., Z. Ding, C. Wang, and Z. Li. "Statistical Analysis of The Patterns and Characteristics of Connected and Autonomous Vehicle Involved Crashes." *Journal of Safety Research* 71 (2019):41-47.

### About the Authors

Hardik Gajera is a Ph.D. student in Civil & Environmental Engineering Department at the University of North Carolina at Charlotte, NC, USA. His current research interests include traffic operations and safety, traffic simulations, and connected and automated vehicles.

**Dr. Srinivas S. Pulugurtha, P.E., F.ASCE** is currently working as a Professor & Research Director of the Department of Civil and Environmental Engineering at the University of North Carolina at Charlotte. He is also the Director of the Infrastructure, Design, Environment, and Sustainability (IDEAS) Center at the University of North Carolina at Charlotte.

**Sonu Mathew** received his Ph.D. degree in Infrastructure and Environmental Systems from the University of North Carolina at Charlotte, NC, USA. He is a Postdoctoral Researcher with Infrastructure, Design, Environment, and Sustainability (IDEAS) Center at the University of North Carolina at Charlotte, NC, USA. His current research interests include traffic safety, traffic flow modeling and simulation, ITS, connected and automated vehicles, and spatial modeling.

### MTI FOUNDER

### Hon. Norman Y. Mineta

### MTI BOARD OF TRUSTEES=

Founder, Honorable Norman Mineta\* Secretary (ret.), US Department of Transportation

Chair, Will Kempton Retired Transportation Executive

Vice Chair, Jeff Morales Managing Principal InfraStrategies, LLC

Executive Director, Karen Philbrick, PhD\* Mineta Transportation InstituteSan José State University

Winsome Bowen Transportation Executive

David Castagnetti Co-Founder Mehlman Castagnetti Rosen & Thomas

Maria Cino Vice President, America & U.S. Government Relations Hewlett-Packard Enterprise

Grace Crunican\*\* Owner Crunican LLC

Donna DeMartino Managing Director Los Angeles-San Diego-San Luis Obispo Rail Corridor Agency John Flaherty Senior Fellow Silicon Valley American Leadership Forum

Stephen J. Gardner \* President & CEO Amtrak

Rose Guilbault Board Member Peninsula Corridor Joint Power Board

Kyle Holland Senior Director, Special Projects, TAP Technologies, Los Angeles County Metropolitan Transportation Authority (LA Metro)

Ian Jefferies\* President & CEO Association of American Railroads

Diane Woodend Jones Principal & Chair of Board Lea & Elliott, Inc.

Steven Keck\* Acting Director California Department of Transportation (Caltrans)

Therese McMillan Executive Director Metropolitan Transportation Commission (MTC)

Abbas Mohaddes President & COO Econolite Group Inc.

Stephen Morrissey Vice President – Regulatory and Policy United Airlines Dan Moshavi, PhD\* Dean Lucas College and GraduateSchool of Business, San José State University

Toks Omishakin\* Secretary California State Transportation Agency (CALSTA)

Takayoshi Oshima Chairman & CEO Allied Telesis, Inc.

Greg Regan President Transportation Trades Department, AFL-CIO

Paul Skoutelas\* President & CEO American Public Transportation Association (APTA)

Kimberly Slaughter CEO Systra USA

Beverley Swaim-Staley President Union Station Redevelopment Corporation

Jim Tymon\* Executive Director American Association of State Highway and Transportation Officials (AASHTO)

\* = Ex-Officio \*\* = Past Chair, Board of Trustees

#### Directors

Karen Philbrick, PhD Executive Director

Hilary Nixon, PhD Deputy Executive Director

Asha Weinstein Agrawal, PhD Education Director National Transportation Finance Center Director

Brian Michael Jenkins National Transportation Security Center Director

